Author's response to reviews

Title: Efficacy, predictability, and safety of small incision lenticule extraction: prospective cohort study

Authors:

   Jae Ryun Kim (oindiano99@gmail.com)
   Hyung Bin Hwang (leoanzel@catholic.ac.kr)
   Su Joung Mun (moons-6@daum.net)
   Young Taek Chung (eyegreen@eyegreen.com)
   Hyun Seung Kim (Sara514@catholic.ac.kr)

Version: 4
Date: 18 September 2014

Author's response to reviews: see over
September 18, 2014

Editorial team

Editorial Office ‘BMC Ophthalmology’

Dear Editorial team of the BMC Ophthalmology

We heartily thank you the editor and referees of the BMC Ophthalmology for taking their valuable time to review our article. We have made some corrections and clarifications in the manuscript after going over the referee’s comments. The changes are summarized below:

Corrections:

1. Title was corrected.

“Efficacy, predictability, and safety of small incision lenticule extraction: 6-months prospective cohort study”

2. Following statement was added in the Methods section

“Written informed consent for participation in the study was obtained from all participants.”

We hope the revised manuscript will better meet the requirements of the BMC Ophthalmology for publication. We thank earnestly you once again for the valuable and constructive review by the referees. We look forward to hearing your response again.

Sincerely yours,

Hyun Seung Kim, M.D.

Professor